Author’s response to reviews

Title: Patients with Glycogen Storage Diseases Undergoing Anesthesia: A Case Series

Authors:

Carmelina Gurrieri (Gurrieri.Carmelina@mayo.edu)
Juraj Sprung (sprung.juraj@mayo.edu)
Toby Weingarten (weingarten.toby@mayo.edu)
Mary Warner (warner.mary@mayo.edu)

Version: 3 Date: 19 Sep 2017

Author’s response to reviews:

BANE-D-17-00182R2

Patients with Glycogen Storage Diseases Undergoing Anesthesia: A Case Series Carmelina Gurrieri, MD; Juraj Sprung, MD, PhD; Toby Weingarten; Mary E Warner, MD BMC Anesthesiology

Dear Dr. Tian,

Thank you for your provisional acceptance of our manuscript "Patients with Glycogen Storage Diseases Undergoing Anesthesia: A Case Series" (BANE-D-17-00182R2). Per Editor request, we now code sex as 1 or 2, and provide age ranges instead of direct ages. This was done for both Table 1 and 2.

Please let us know if there are any other edits that are required.

Sincerely yours,

Toby Weingarten